Author’s response to reviews

Title: Predictive multi-imaging biomarkers relevant for visual acuity in idiopathic macular telangiectasis type 1

Authors:

Jingli Guo (18553522773@163.com)
Wenyi Tang (drtangwy@163.com)
Xiaofeng Ye (newnewye@163.com)
Haixiang Wu (whx577@163.com)
Gezhi Xu (drxugezhi@163.com)
Wei Liu (bfgf14@aliyun.com)
Yongjin Zhang (yongjinzhang@yahoo.com)

Version: 1 Date: 15 Nov 2017

Author’s response to reviews:

1. We note that author ‘Gezhi Xu’ entered in the editorial submission system is not included in the list of authors on the manuscript title page. Please amend the title page of your manuscript to state the names of all authors in full.

Answer : Thank you for your reminding. We have added all detailed information of every author.

2. Please include an email address for each author on the title page.

Answer We have written specific email of every author on the title page.
3. Please confirm in your ethics statement whether the participant consent was written or verbal. If it was verbal, please confirm that you received ethics committee approval for verbal consent and detail how you documented the consent. If the need for informed consent has been waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the Board or a reference to the relevant legislation. Please add your updated ethics statement to the relevant section under the Declarations subheading.

Answer: This research was a respective study. All the patients volunteered for the study in a verbal manner by telephone when our team began to collect clinical data. We have already received ethics committee approval for verbal consent.

4. Please amend your Author Contributions to clearly state the involvement of all authors.

Answer: GJL: made substantial contributions to conception and design, acquisition of data, analysis and interpretation of data; TWY: made substantial contributions to analyse and interpret the data; YXF: acquisition of data and research design; WHX: acquisition of data and research design; XGZ: Research design. LW: given final approval of the version to be published, research design and manuscript preparation. ZYJ: given final approval of the version to be published, research design, data analysis, and manuscript preparation. All authors read and approved the final manuscript.